

Dear reviewers and editors:

Thank you very much for your kind consideration for publication of our manuscript entitled “Neurogenic orthostatic hypotension with Parkinson's disease as a cause of syncope: a case report”. On behalf of my co-authors, we would like to express our great appreciation to you and reviewers. According to the comments and suggestions of reviewers and editor, we have revised the manuscript, and responded point by point, the responses are listed below. I would like to re-submit this revised manuscript, and hope it is acceptable for publication.

Reviewer #1:

Scientific Quality: Grade C (Good)

Language Quality: Grade B (Minor language polishing)

Conclusion: Minor revision

Specific Comments to Authors: Greetings, I attach some small suggestions 80 three times a day) and Slangeline (5 mg once a day). - Correct is Selegiline 82 There was no significant personal or family history to note - It was mentioned the Parkinson’s disease since 6 monthsago - How old is the high blood pressure? 119 OUTCOME AND FOLLOW-UP 120 After treatment, dizziness and amaurosis gradually relieved, and blood pressure became much 121 more stable. At the 3 month follow-up, there was no recurrence syncope. - Please attach a 24-hour ambulatory blood pressure monitoring after treatment (to objectivate the affirmation)

Answer:

We have reviewed our paper throughout and corrected some **Language Errors** and make sure there is no other mistake.

Comment 1: 80 three times a day) and Slangeline (5 mg once a day)- - Correct is Selegiline

Thank you for your suggestion. We have placed Selegiline with Slangeline in the paper.

Comment 2: 82 There was no significant personal or family history to note- It was mentioned the Parkinson’s disease since 6 months ago - How old is the high blood pressure?

We described the history of Parkinson’s disease and the hypertension in the part of history of past illness.

Comment 3: 119 OUTCOME AND FOLLOW-UP 120 After treatment, dizziness and amaurosis gradually relieved, and blood pressure became much 121 more stable. At the 3 month follow-up, there was no recurrence syncope. - Please attach a 24-hour ambulatory blood pressure monitoring after treatment (to objectivate the affirmation)

It is really a valuable suggestion. We did a telephone follow up in the last 2 week and suggested the patient to recheck 24-hour ambulatory blood pressure monitoring. Luckily, the patient adopted the opinion, and the 24-hour ambulatory blood pressure monitoring indicated that blood pressure was much more stable compared with that 8 month ago. We have added this results in OUTCOME AND FOLLOW-UP, Table 1and Figure 1.

EDITORIAL OFFICE’S COMMENTS

(1) Science editor: 1 **Scientific quality:** The manuscript describes a case report of the neurogenic orthostatic hypotension with Parkinson's disease as a cause of syncope. The topic is within the scope of the WJCC. (1) **Classification:** Grade C; (2) **Summary of the Peer-Review Report:** This is a good case report. The questions raised by the reviewers should be answered; (3) **Format:** There is 1 table and 2 figures; (4) **References:** A total of 21 references are cited, including 5 references published in the last 3

years; (5) Self-cited references: There is no self-cited reference; and (6) References recommendations: The authors have the right to refuse to cite improper references recommended by the peer reviewer(s), especially references published by the peer reviewer(s) him/herself (themselves). If the authors find the peer reviewer(s) request for the authors to cite improper references published by him/herself (themselves), please send the peer reviewer's ID number to editorialoffice@wjgnet.com. The Editorial Office will close and remove the peer reviewer from the F6Publishing system immediately. 2 Language evaluation: Classification: Grade B. A language editing certificate issued by EditSpring was provided. 3 Academic norms and rules: The authors provided the written informed consent. No academic misconduct was found in the Bing search. 4 Supplementary comments: This is an unsolicited manuscript. No financial support was obtained for the study. The topic has not previously been published in the WJCC. 5 Issues raised: (1) The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor. 6 Recommendation: Conditional acceptance.

Answer:

We have uploaded original figure documents using PowerPoint in the F6Publishing system, and also edited the citation according to the required form. We have reviewed our paper throughout and the language has been polished by EditSprings once again.

(2) Editorial office director:

(3) Company editor-in-chief: I have reviewed the Peer-Review Report, the full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the World Journal of Clinical Cases, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office's comments and the Criteria for Manuscript Revision by Authors. Before its final acceptance, the author(s) must provide the Signed Informed Consent Form(s) or Document(s). For example, authors from China should upload the Chinese version of the document, authors from Italy should upload the Italian version of the document, authors from Germany should upload the Deutsch version of the document, and authors from the United States and the United Kingdom should upload the English version of the document, etc.

Answer:

We have uploaded the Chinese version of Signed Informed Consent Form in the F6Publishing system.

I would be happy to make any further changes that may be required. Best regards!

Ya Li

3318007@zju.edu.cn